Author’s response to reviews

Title: Guideline use among different healthcare professionals in diagnosing Attention Deficit Hyperactivity Disorder in Dutch children; who cares?

Authors:

Birgit Levelink (birgit.levelink@mumc.nl)
Lonneke Walraven (lonnekewalraven@gmail.com)
Edward Dompeling (edward.dompeling@mumc.nl)
Frans Feron (f.feron@maastrichtuniversity.nl)
Dorothea van Zeben-van der Aa (thea.van.zeben@mumc.nl)

Version: 2 Date: 19 Oct 2018

Author’s response to reviews:

Thank you for your comments. I will answer them point to point.

1. Please replace the term “Attention-Deficit Hyperactive Disorder” with “Attention Deficit Hyperactivity Disorder” throughout the manuscript (eg, Background, Row 7 Page 6), I adjusted this term.

2. In the Background paragraph, you should provide a more comprehensive description of all the guidelines developed in the Netherlands for the diagnosis and management of ADHD, by reporting their key elements and by highlighting in what they are similar and different to each other. This can help the reader understanding the Dutch context; I provided a description of the guidelines and their similarities which are important for the rest of the artikel.

3. Page 6 Row 50-55. Please specify the similarities of the Dutch guidelines with those developed by the NICE; I did not add extra text here, reading the added text concerning the dutch guidelines (comment 2) also explains this point.